Author's response to reviews

Title: P. marneffei Infection Involves the Primary Tracheal Structure: the First Case Report

Authors:

Ye Qiu (yegiu2013graduated@163.com)
Jianquan Zhang (jqzhang2002@sina.com)
Guangnan Liu (liguoqiuye@gmail.com)
Xiaoning Zhong (liguoqiuye@gmail.com)
Zhiyi He (liguoqiuye@gmail.com)
Jingmin Deng (liguoqiuye@gmail.com)
Jing Bai (liguoqiuye@gmail.com)

Version: 2
Date: 27 December 2013

Author's response to reviews: see over
Dear Editor Neil,

Thank you for contacting me for some changes. I have made all changes you want.

This is a manuscript entitled “P. marneffei Infection Involves the Primary Tracheal Structure: the First Case Report”. It is submitted to be considered for publication as a “Brief report” in your journal. This paper is new. Neither the entire paper nor any part of its content has been published or has been accepted elsewhere. It is not being submitted to any other journal. All authors have seen the manuscript and approved to submit to your journal.

**Requesting consent statement (case report):**

Written informed consent was obtained from the patient for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor of this journal.

Thank you very much for your attention and consideration.

**Competing interests**

All the authors declare that they have no competing interests.

**Authors’ contributions**
YQ have made substantial contributions to conception and design, acquisition of data, analysis and interpretation of data and have been involved in drafting the manuscript. JZ and GL have made substantial contributions to conception and design, acquisition of data, analysis and interpretation of data, have been involved in drafting the manuscript, revising it critically for important intellectual content, have given final approval of the version to be published and agree to be accountable for all aspects of the work in ensuring that questions related to the accuracy and integrity of any part of the work are appropriately investigated and resolved. XZ have given final approval of the version to be published. ZH have analysis and interpretation of data and agree to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved. JD and JB conceived of the study, and participated in its design and helped to draft the manuscript. All authors read and approved the final manuscript.

Please do not hesitate to contact us if you have any questions.
Best wishes.

Yours sincerely,

Jianquan Zhang MD, PhD

Department of Respiratory Medicine, the First Affiliated Hospital of Guangxi Medical University, Nanning, Guangxi, 530021, China

Tel.: +86 13978123845

E-Mail: jqzhang2002@sina.com